Author's response to reviews

Title: Self-rated Health among Mayan Women participating in a Randomised Intervention Trial reducing Indoor Air Pollution in Guatemala

Authors:

Esperanza Diaz (esperanza.diaz@isf.uib.no)
Nigel Bruce (nrb@liverpool.ac.uk)
Dan Pope (danpope@liverpool.ac.uk)
Anaite Diaz (anaite_artiga@yahoo.com)
Kirk R Smith (krrksmith@berkeley.edu)
Tone Smith-Sivertsen (tone.smith-sivertsen@isf.uib.no)

Version: 2 Date: 10 March 2008

Author's response to reviews:

Dear Editor:

Thank you very much for your email. Our trial did not have any registration number, as it began before this number was necessary.

Asking to your second question, the consent in the study was oral, as most of the women were illiterate. This information is now included in the text.

Kind regards

Esperanza Diaz